

Working Paper Series

Carlo Altavilla, Miguel Boucinha, Marco Pagano, Andrea Polo Climate risk, bank lending and monetary policy



Disclaimer: This paper should not be reported as representing the views of the European Central Bank (ECB). The views expressed are those of the authors and do not necessarily reflect those of the ECB.

Abstract

Combining euro-area credit register and carbon emission data, we provide evidence of a climate risk-taking channel in banks' lending policies. Banks charge higher interest rates to firms featuring greater carbon emissions, and lower rates to firms committing to lower emissions, controlling for their probability of default. Both effects are larger for banks committed to decarbonization. Consistently with the risk-taking channel of monetary policy, tighter policy induces banks to increase both credit risk premia and carbon emission premia, and reduce lending to high emission firms more than to low emission ones. While restrictive monetary policy increases the cost of credit and reduces lending to all firms, its contractionary effect is milder for firms with low emissions and those that commit to decarbonization.

JEL classification: E52, G21, Q52, Q53, Q54, Q58.

Keywords: climate risk, carbon emissions, interest rate, lending, monetary policy.

Non-Technical Summary

Do banks penalise climate risk in their lending activity? If so, does this apply particularly to banks publicly committed to low-carbon targets? Does monetary policy affect the climate risk premium required by banks? These are key issues for policymaking, considering that the ECB is committed to taking climate risk into account in its supervisory framework and climate change in its monetary policy framework. However, there is no consensus in the literature on these issues. For example, existing evidence on whether banks price climate risk in their lending policies is far less clear-cut than that regarding the pricing of climate risk in bond and stock markets. Views also differ on whether the disclosure made by banks on their commitment to decarbonisation is reliable. Another key question which is until now totally unexplored is whether changes in monetary policy affect the climate risk premia charged by banks to polluting firms. This paper addresses these questions by combining granular credit register data for euro area banks and firms' carbon emission data between September 2018 and December 2022.

Whether banks should care about climate risk in their lending policies is not obvious. In principle, they should care about climate risk only as long as it affects their clients' default risk. For instance, a bank lending to an oil company should consider the risk that carbon taxes or environmental regulations may end up forcing the company into default. However, the models used by banks to assess credit risk may not fully capture tail-risk events such as future changes in regulation or technology. Moreover, these models may fail to capture systemic risk arising from adverse environmental shocks or unforeseen changes in carbon taxes. Last but not least, banks may internalise social concerns about climate risk, in response to pressure from the media, depositors and activist shareholders.

Simple descriptive statistics suggest that during our sample period euro area banks price climate risk: the monthly mean interest rate charged to firms in the top quartile by current carbon emissions consistently exceeds that charged to firms in the bottom quartile. The difference between the two is about 14 basis points, on average. Moreover, the rate charged to firms that have not committed to reducing future emissions consistently exceeds that charged to committed firms, with the overall difference averaging 20 basis points. Hence, banks also appear to differentiate their lending rates based on their clients' prospective carbon emissions, not just their current ones.

Empirical analysis based on panel regressions confirms this result: controlling for firms' probability of default, euro area banks charge a higher interest rate to firms with higher current carbon emissions and a lower rate to those that commit to reducing their emissions in the future. These results are true also within clusters of firms in the same industry, location, and size class and even when controlling for any time-invariant firm characteristic. We also find that banks that publicly commit to low-carbon targets lend consistently with such commitment: committed banks charge a higher climate risk premium to high-emission firms and give a larger discount to firms that commit to lowering their future emissions.

Our second research question is whether changes in monetary policy affect the climate risk premia that banks charge to polluting firms. This is an even less obvious question and is still unexplored. Two standard models of the effects of monetary policy on financial intermediation yield opposite predictions on this point.

On one hand, models of the risk-taking channel of monetary policy can be extended to encompass climate risk. These models predict that expansionary monetary policy induces banks to reduce monitoring efforts and lend to riskier borrowers, engaging in yieldseeking policies, while restrictive monetary policy has the opposite effect. Extending these models to climate risk, a monetary expansion should lead banks to lower the climate risk premium they charge to high-emission firms relative to low-emission ones, and a monetary restriction should induce banks to raise this premium. In this framework, a monetary easing would favour relatively more brown firms through a 'climate risk-taking channel'. Symmetrically, a monetary tightening would constrain lending conditions relatively less for green firms.

On the other hand, in the wake of a monetary tightening, banks may be more inclined to restrict lending to firms with innovative, hence lesser known, low-carbon technologies than to firms with traditional, carbon-intensive technologies. Moreover, the intangible nature of innovation makes the projects of the former harder to collateralize than those of the latter, so carbon-intensive firms tend to have more tangible capital assets than low-carbon firms. Thus, restrictive monetary policy can be expected to curtail lending to low-emission firms more than to comparable high-emission ones, and symmetrically expansionary monetary policy to facilitate lending to the former more than to the latter. Hence, if low-emission firms are more financially constrained than high-emission ones, the contractionary effect of a monetary tightening should be larger for green than brown firms. This is exactly the opposite prediction to that implied by the risk-taking channel of monetary policy.

Our evidence on the effects of monetary policy strongly supports the first view: it uncovers a climate risk-taking channel of monetary policy. Unexpected increases in the ECB's policy rate are associated with an increase in the climate risk premium charged to high-emission firms, over and above the baseline increase in risk premium. Moreover, such restrictive monetary policy shocks are associated with a smaller increase in the climate risk premium charged to firms committed to lowering emissions. Symmetrically, over the subsequent year they result in a significantly greater contraction of lending to high-emission firms than to low-emission ones, and in a lower contraction of lending to firms that commit to a target level of emissions.

These findings are highly relevant to assess the environmental effects of the monetary policy. Of course, the findings do not amount to saying that a monetary restriction is good for the environment: restrictive monetary policy generally tightens financing conditions, and thus will also slow down investments aimed at reducing carbon emissions. However, our evidence suggests that a monetary tightening worsens financing conditions more for high-emission firms, especially those not committed to reducing them. While a restrictive monetary policy stance may slow down the pace of decarbonisation, it should be expected to tighten financing conditions comparatively less for firms that are greener or committed to becoming greener than for other firms.

1 Introduction

We combine granular credit register data for all euro-area banks and carbon emission data to address two related research questions: do banks penalize climate risk in their lending activity? If so, is the climate risk premium that they require affected by monetary policy? Neither one is an obvious question.

In principle, banks should care about their clients' exposure to climate risk only insofar as this affects their default risk: for instance, a bank lending to an oil company should take into account the risk that carbon taxes or environmental regulations may end up forcing the company into default. However, the typical models that banks use to assess credit risk may be unable to capture tail-risk events such as future changes in regulation or in technology. Moreover, these models may fail to capture the systemic risk arising from adverse environmental shocks or unforeseen changes in carbon taxes, and to allow for the resulting concerns of macroprudential regulators. Last but not least, banks may internalize social concerns about climate risk for reputational reasons, in response to pressure from the media, depositors and activist shareholders. However, media attention and investors' activism may be less successful in affecting banks' environmental record than that of mutual or pension funds, because banks' loan portfolios are typically far more opaque than the securities portfolios of institutional investors. This may explain why so far the evidence on whether banks price climate risk in their lending policies is far less clear cut than that regarding the pricing of climate risk in security markets.

Our evidence is that euro-area banks not only price the current climate risk exposure of the firms they lend to, but also their future exposure: they charge a higher interest rate to firms with higher current carbon emissions, controlling for standard sources of credit risk, but a lower rate to those that commit to reduce their emissions in the future. Moreover, we find that banks that publicly commit to environmentally responsible lending policies charge a higher climate risk premium to high-emission firms and give a larger discount to firms that commit to lower their future emissions. Whether changes in monetary policy affect the climate risk premia charged by banks on polluting firms is an even less obvious question, and in fact a still unexplored one. Two standard models of the effects of monetary policy on financial intermediation yield opposite predictions on this issue.

On one hand, models of the risk-taking channel of monetary policy can be extended to encompass climate risk. Recall that these models predict that expansionary monetary policy induces banks to reduce monitoring efforts and lend to riskier borrowers, engaging in yield-seeking policies, while restrictive monetary policy has the opposite effect (Acharya and Naqvi, 2012; Adrian and Shin, 2009; Borio and Zhu, 2012; Dell'Ariccia, Laeven, and Marquez, 2014). Extending these models to climate risk, a monetary expansion would comparatively favor high-emission firms, and symmetrically a monetary restriction would comparatively favor low-emission ones.

On the other hand, banks may be more inclined to restrict lending to firms with innovative, lesser known low-carbon technologies than to firms with traditional carbonintensive technologies. Moreover, the intangible nature of innovation makes the projects of the former harder to collateralize than those of the latter: indeed carbon-intensive firms tend to have more tangible capital assets than low-carbon firms (Iovino, Martin, and Sauvagnat, 2023). Thus, insofar as monetary policy disproportionately affects lending to credit-constrained firms (Bernanke and Gertler, 1989, 1995), one may expect monetary restrictions to curtail the debt capacity of low-emission firms more than that of comparable high-emission ones, and symmetrically monetary expansions to facilitate lending to the former more than to the latter. Hence, insofar as low-emission firms are more financially constrained, monetary policy changes should affect them more than high-emission firms, which is exactly the opposite prediction to that implied by the risk-taking channel of monetary policy.

Our evidence on the effects of monetary policy strongly supports the first of these two predictions: it uncovers a "climate risk-taking channel of monetary policy". Unexpected increases in the ECB's policy rate are associated with an increase in the climate risk premium charged to high-emission firms, over and above the increase in risk premium charged to firms with high probability of default. Moreover, such restrictive monetary policy shocks are associated with smaller increases in the climate risk premium charged to firms that are committed to lower their emissions.¹

Symmetrically, over the subsequent year restrictive monetary policy results in a significantly greater contraction of lending to high emission firms than to low emission ones, and in a lower contraction of lending to firms that commit to a target level of emissions. The fact that carbon-intensive firms experience *both* stronger increases in risk premia and sharper drops in lending after restrictive monetary policy shocks suggests that these shocks curtail the *supply* of credit to these firms more than to low-emission firms (rather than affecting differently their demand for credit). The same applies to the finding that firms committed to an emission target feature *both* lower increases in risk premia and milder drops in lending than uncommitted firms after restrictive monetary policy shocks.

Importantly, our findings do not amount to saying that a monetary restriction is good for the environment: insofar as restrictive monetary policy generally tightens financing conditions and slows down investment, it will also slow down investments aimed at reducing carbon emissions (Levine, Lin, Wang, and Xie, 2018). But our evidence suggests that a monetary tightening worsens financing conditions more for high-emission firms and for those that do not commit to reduce them. These findings are quite relevant in view of recent concerns that "monetary policy tightening may ultimately slow down the pace of decarbonisation".²

¹We identify monetary policy shocks using high-frequency monetary policy surprises based on the Euro Area Monetary Policy Event-Study Database (EA-MPD) developed by Altavilla, Brugnolini, Gürkaynak, Motto, and Ragusa (2019). Results are robust if we use dummies for the change in monetary policy stance in 2022, as captured either by the first rate hike in July 2022 or by the quantitative tightening announcement in December 2021, which may have signaled subsequent policy rate normalisation).

² "Monetary policy tightening and the green transition," speech by Isabel Schnabel, Member of the Executive Board of the ECB, Stockholm, 10 January 2023. The concern that monetary policy may have undesirable effects on climate risk also transpires from the ECB's statement that it considers climate risk for its corporate sector asset purchases and will include climate change considerations in its monetary policy strategy (press release of 8 July 2021). This is particularly relevant in view of fact that "brown firms" usually issue relatively more bonds, so that central bank open market purchases designed to be market-neutral actually end up channeling more funds to "brown firms" (Papoutsi, Piazzesi, and Schneider, 2022).

The rest of the paper proceeds as follows. Section 2 places the paper against the backdrop of recent research on the pricing of climate risk and on the effects of monetary policy on the pricing of risk. Section 3 describes the data. Sections 4 and 5 present the empirical analysis. Section 6 concludes.

2 Literature

This paper contributes to two distinct, and so far disconnected, strands of research: on one hand, the rapidly growing literature on the pricing of climate risk in financial markets and more specifically in credit markets; on the other hand, the research on the effect of monetary policy on risk taking by banks.

Due to the growing concern about global warming, climate risk has started playing a growing role in the valuation of financial assets. Climate risk is generally broken down into two components: physical risk, which refers to the economic harm caused by natural hazards, such as floods and wildfires, and transition risk, stemming from regulatory changes aimed at reducing emissions and facilitating the transition towards a greener economy. Most empirical studies have analyzed the asset pricing implications of transition risk, using different measures of carbon intensity or environmental friendliness as proxies of firm exposure to climate risk.

In equity markets, there is evidence that firms with higher carbon emissions are valued at a discount (Bolton and Kacperczyk, 2021, 2022; Bolton, Halem, and Kacperczyk, 2022), high-pollution intensity firms pay larger average annual returns than low-pollution firms in the same industry (Hsu, Li, and Tsou, 2023), and firms with high environmental scores have higher returns at times of negative news about future climate change (Engle, Giglio, Kelly, Lee, and Stroebel, 2020). In option markets, the cost of hedging extreme downside risks is larger for more carbon-intensive firms, especially at times of heightened public attention to climate risk (Ilhan, Sautner, and Vilkov, 2020). Transition risk is also priced in fixed income markets: corporate bonds that perform better at times of bad news about climate change, hence are less exposed to climate risk, pay lower returns (Huynh and Xia, 2021), and "green bonds", whose proceeds are expressly tied to environmentally friendly projects, trade at lower yields than bonds with similar characteristics but without a green designation according to Baker, Bergstresser, Serafeim, and Wurgler (2022), although Flammer (2021) reports no such difference in yields.

For credit markets, instead, so far the evidence on the pricing of transition risk is far less clear cut and only refers to the syndicated loan market, which only represents a segment of the credit market, mostly catering to the largest companies. According to Beyene, De Greiff, Delis, and Ongena (2021), banks in the syndicated loan market do not significantly price the risk stemming from the potential stranded assets held by fossil fuel firms, while bond markets price this risk. Ehlers, Packer, and De Greiff (2022) instead find that, following the 2015 Paris Agreement, polluting firms started paying a "carbon premium" on their syndicated loans.³ Our evidence addresses this issue by relying on panel data for the entire euro-area credit market, rather than for syndicated loans only, which in Europe are around 10 per cent of all loans, and reveals that banks not only price their loans based on firms' current carbon emissions, but also based on firms' commitments to reduce future emissions.

While the Paris Agreement binds entire countries, other initiatives such as the Net-Zero Banking Alliance and the United Nations' Environmental Programme Finance Initiative, have directly addressed financial intermediaries, committing them to environmentfriendly financing policies. Whether banks abide by these commitments or not is also controversial: Kacperczyk and Peydro (2021) find that banks adhering to the Science Based Targets initiative (SBTi) honored their commitment by lending less to high-emission firms in the syndicated loan market, and Degryse, Goncharenko, Theunisz, and Vadasz (2020) document that after 2015 "green banks", i.e., those that commit to lend preferentially to low emission companies, reward "green firms" by offering them cheaper syndicated

³There is also evidence that banks price physical climate risk: firms in locations with higher exposure to climate change pay higher spreads on their bank loans (Javadi and Al Masum, 2021), and so do borrowers with collateral consisting of properties exposed to a greater risk of sea level rise (Nguyen, Ongena, Qi, and Sila, 2022).

loans.⁴ However, Ehlers, Packer, and De Greiff (2022) find that "green banks" do not price carbon risk differently from other banks, and Giannetti, Jasova, Loumioti, and Mendicino (2023) find that banks that emphasize the sustainability of their lending policies in their investment reports in fact lend more to "brown firms" and do not provide more credit to firms in green industries. Our own evidence indicates that SBTi signatories not only require greater loan premia from high emission firms, but also charge lower premia to those committing to emission-reducing investments.

In the aftermath of the Paris climate agreement, several governments enacted environmental regulations, such as carbon taxes. Ivanov, Kruttli, and Watugala (2023) exploit the design of the cap-and-trade bill in California to investigate its effects on banks' lending policies, and find that, since its enactment, polluting firms face shorter maturities and have lower access to bank financing. However, the effect of domestic environmental policies may be reduced by banks' tendency to exploit more lenient regulations in other countries: Benincasa, Kabas, and Ongena (2022) find that, upon facing more stringent regulations in their home country, banks increase cross-border lending to firms located in countries with lighter policies; similarly, Laeven and Popov (2022) find that the introduction of a carbon tax is associated with an increase in domestic banks' lending to coal, oil, and gas companies in foreign countries. But there seem to be important exceptions to such regulatory arbitrage: European banks reduced credit to U.S. polluting firms after President Trump's announcement of withdrawal from the Paris Agreement, possibly for reputational reasons, in response to strong public pressure in favor of environmental policies (Reghezza, Altunbas, Marques-Ibanez, Rodriguez d'Acri, and Spaggiari, 2022).

The second strand of research to which we contribute is that on the risk-taking channel of monetary policy: when monetary policy is expansionary, lending standards become softer, particularly for riskier borrowers, so that banks' loan portfolios become riskier; conversely, when monetary policy becomes tighter, banks raise their lending standards and de-risk their loan portfolios. Incentive problems within banks may be at the origin of

 $^{^{4}}$ Houston and Shan (2021) also find assortative matching of banks and firms based on their respective ESG scores.

such changes in their risk-taking behavior. Acharya and Naqvi (2012) propose a model where banks elicit effort from loan officers by tying their compensation to the volume of loans: in the presence of abundant liquidity, volume-based compensation induces greater risk taking, lowers the sensitivity of bankers' payoffs to downside risks and induces excessive lending. Relatedly, Dell'Ariccia, Laeven, and Marquez (2014) propose a model where banks can reduce the credit risk of their loan portfolio via costly monitoring, and show that if they can adjust their capital structure in response to interest rate shocks, in equilibrium they will respond to a drop in risk-free rates by increasing leverage, reducing monitoring and increasing exposure to risk. Gambacorta (2009) points out that the risktaking channel of monetary policy may not only result from banks' greater yield-seeking incentives when monetary policy is expansionary, but also from the impact of low interest rates on the value of firms' assets and cash flows, which can in turn can affect banks' valuation of their default risk.

Much evidence is consistent with the risk-taking channel of monetary policy. Using U.S. data, Dell'Ariccia, Laeven, and Suarez (2017) document that ex ante risk-taking by banks (measured by banks' internal risk rating of new loans) is negatively associated with increases in short-term interest rates. Using very granular Spanish credit register data, Jiménez, Ongena, Peydró, and Saurina (2014) show that a lower overnight interest rate induces less capitalized banks to accept more loan applications by ex-ante risky firms and to lend them more with fewer collateral requirements, although they have a higher ex-post likelihood of default. Anderson and Cesa-Bianchi (2023) investigate whether monetary policy surprises based on high-frequency data have different impact on firms with different leverage, hence different ex-ante risk, and find that high leverage firms experience a more pronounced increase in credit spreads than firms with low leverage in response to a monetary policy tightening, and that most of this increase reflects higher risk premia charged by banks, rather than a revision of the firms' expected default risk.

While we also investigate the differential impact of an unexpected monetary tightening on firms' credit spreads, our focus is on its differential effect on firms arising both from their credit risk (as measured by their PD) and from their climate risk (as captured by their current and planned carbon emissions), rather than from different leverage, as done by Anderson and Cesa-Bianchi (2023). As such, our approach is based on an extended version of the risk-taking channel of monetary policy, where changes in monetary policy should also affect banks' pricing of climate risk: restrictive monetary policy should induce banks to penalize high-emission more than comparable low-emission firms, while expansionary monetary policy should induce banks to favor the former more than the latter. In both cases, the effect of monetary policy shocks on interest rates charged to more polluting firms should be mitigated if these credibly commit to investments aimed to reduce their carbon footprint, as this would reduce their future exposure to climate risk.

Interestingly, our finding that restrictive monetary policy shocks raise the cost of credit to high-emission firms more than to low-emission firms dovetails with the result by Döttling and Lam (2024) that the stock prices of high-emission U.S. firms drop more than those of low-emission ones in response to restrictive monetary policy shocks around Federal Open Market Committee announcements. Hence, in line with our findings, they find that monetary policy has a stronger effect on the cost of capital of firms more exposed to climate transition risk.

3 Data

We draw loan-level information obtained from AnaCredit, a proprietary and confidential database of the ECB and the national central banks of euro-area countries (the Eurosystem). AnaCredit is a granular (transaction-level) database that reports 94 loan-level attributes on a monthly frequency in a harmonised way for all euro-area countries. The minimum reporting threshold for loans to firms is set at $\pounds 25,000$ for all countries participating in the database.

AnaCredit covers a comprehensive set of credit instruments: overdrafts, revolving

credit, credit lines, reverse repurchase agreements, and other loans, including term loans.⁵ For each instrument, it reports the interest rate charged by the issuing bank and its estimate of the probability of default (PD), i.e., the likelihood that the borrower will not make scheduled repayments over a one-year horizon. The sample period ranges from September 2018 (the first available month in AnaCredit) to December 2022. The descriptive statistics shown in Table 1 show that our key variable of interest, i.e., the monthly interest spread charged by banks on their loans over the contemporaneous duration-matched risk-free rate, averages to 151 basis points, with a standard deviation of 76 basis points, ranging from 54 basis points for firms in the lowest decile to 241 basis points for those in the top decile.⁶ The PD is defined at the firm-time level as a weighted average of the estimate reported by each bank lending to the firm. Compared to the spread, it has a somewhat more right-skewed cross-sectional distribution, with above-mean PD values being concentrated in the top decile.

For each firm in the euro area, we merge AnaCredit data regarding its credit relationships with all its lenders with data about the firm's current carbon emissions and its planned carbon reduction targets, both drawn from the Refinitiv database. The reliance on emission data confines our sample to publicly listed firms in AnaCredit. This is because emission data is available for these firms, which are subject to stricter reporting standards. However, given that large companies generate the most emissions and account for the largest share of loans, they are the biggest source of transition risk for the banking sector (Alogoskoufis, Dunz, Emambakhsh, Hennig, Kaijser, Kouratzoglou, Muñoz, Parisi, and Salleo, 2021).⁷ Under the Greenhouse Gas (GHG) protocol, a firm's GHG emissions are classified in three categories for accounting and reporting purposes: Scope 1 emissions

 $^{^{5}}$ The complete list of instruments also includes credit card debt, trade receivables, financial leases and deposits other than reverse repurchase agreements.

 $^{^{6}}$ As an example, for a 5-year fixed rate loan the spread is computed relative to the 5-year OIS, whereas for a loan of the same maturity with a variable rate resetting every 3 months it is benchmarked against the 3-month OIS rate.

⁷According to Alogoskoufis, Dunz, Emambakhsh, Hennig, Kaijser, Kouratzoglou, Muñoz, Parisi, and Salleo (2021): "Although large firms represent an extremely small portion of the sample in terms of number of companies, they clearly represent the highest share of exposures [to climate risk] for euro area banks. Additionally, large companies seem to be the biggest polluters given that they contribute almost 90% of the overall emissions as against 50% in terms of their total exposures."

are those directly produced by sources owned or controlled by the firm; Scope 2 emissions are those associated with the consumption of purchased energy; Scope 3 emissions include all those that occur in the value chain of the firm, excluding Scope 2 ones. We measure the current emissions of firm f (*Carbon*_{f,t}) as the sum of Scope 1 and Scope 2 emissions of CO2 (and CO2 equivalents) in the previous year (or quarter, depending on the firm's reporting frequency), in thousands of tonnes scaled by net revenues in million US dollars. This transformation removes the obvious bias otherwise arising from large firms featuring higher emissions due to the scale of their operations. We exclude Scope 3 emissions, as this information is less reliable.

Table 1 shows that the distribution of carbon emissions is heavily right-skewed, the mean being 0.18 and the median 0.03: only the top decile features emissions larger than the mean. Hence, insofar as a firm's PD and its current emissions are taken to measure a firm's credit and climate risk, respectively, Table 1 indicates that both types of risk are concentrated in the top 20% of the firms' distribution. Yet, even the emissions of firms in the top decile of our sample (530 tons emissions per million US dollars) is about half the size of emissions by firms in the top decile of the international sample of companies in Ehlers, Packer, and De Greiff (2022).

While the level of emissions refers to firms' current environmental performance, we also consider a forward-looking variable that measures firm's commitment to reduce future emissions, namely, a dummy variable that equals 1 if in month t a given firm f has disclosed an emission reduction target and 0 otherwise $(Target_{f,t})$. Table 1 indicates that 58% of the firm/month observations in our sample refer to firms that have committed to reduce emissions. Disclosing an emission target appears to be a mechanism enabling high-emission firms to signal their plan to reduce their carbon footprint: the emission intensity of committed firms is 0.23, against 0.12 for non-committed firms. Bolton and Kacperczyk (2023) show that indeed firms disclosing an emission target subsequently reduce their emissions, and find that in Europe a greater fraction of high emitters have announced such a target than in North America. We complement these data regarding firms' commitment to reduce emissions with information regarding banks' environmental commitment. Following Kacperczyk and Peydro (2021), we identify the banks that signed a commitment letter in the context of the Science Based Targets initiative (SBTi). The SBTi is a joint initiative by Carbon Disclosing Project (CDP), the UN Global Compact, the World Wide Fund for Nature (WWF), and the World Resources Institute (WRI), whose purpose is to define and promote net-zero targets in line with the climate science. The overall goal of the initiative is to induce companies to commit to decarbonization pathways, so as to increase the chance that global emissions are reduced to a level that keeps the rise in average temperature below 1.5°C. To join the SBTi, a company must first sign a commitment letter stating that it will work to set a science-based emission reduction target.

The Refinitiv database provides information about whether at date t a given bank b is a signatory of the SBTi letter or not $(Commit_{b,t})$. Table 1 shows that only 11% of the bank/month observations in our sample refer to banks that have signed the SBTi letter. A small number of non-financial companies have also had their target reduction emission validated by STBi, but Carbone, Giuzio, Kapadia, Krämer, Nyholm, and Vozian (2021) find very similar patterns of emission reductions for firms with an SBTi verified target and those with a self-disclosed emission reduction target only, based on Refinitiv: we rely on the latter, since it dominates the former in terms of data coverage.

Finally, as a measure of monetary policy shocks, we use high-frequency monetary policy surprises based on the Euro Area Monetary Policy Event-Study Database (EA-MPD) developed by Altavilla, Brugnolini, Gürkaynak, Motto, and Ragusa (2019). These surprises are calculated by measuring changes in risk-free rates in a narrow time window around official monetary policy announcements. More precisely, for each Governing Council meeting, the realised policy surprise (MP_t) is measured as the change in interest rates from 15 minutes before the press release to 15 minutes after the press conference. We use the "target" factor as defined in Altavilla, Brugnolini, Gürkaynak, Motto, and Ragusa (2019), with estimated loadings reaching the maximum at very short maturities (1-3 months) and monotonically decreasing across the maturity spectrum. In other words, this factor summarises the footprint of a policy rate shock on the term structure. Table 1 reveals that in our sample period the median monetary policy surprise is zero, but the mean is positive (1.09 basis points), due to the tightening announcements that occurred since July 2022. Such high-frequency measures of monetary policy shocks, pioneered by Gurkaynak, Sack, and Swanson (2005), have recently been used by Jarociński and Karadi (2020) and Anderson and Cesa-Bianchi (2023), among others.

4 Do Banks Price Climate Risk?

In this section, we use the euro-area data described in Section 3 to answer two related questions. First, do banks price climate risk in their lending rates? Second, do committed banks place a higher price on climate risk? We leave the analysis of the impact of monetary policy on banks' pricing of climate risk to Section 5.

4.1 Banks' pricing of climate risk

We start by providing descriptive evidence on banks' pricing of climate risk, by plotting the interest rates charged by the banks in our sample to firms with different carbon footprints between September 2018 and December 2022. Figure 1 plots monthly values of the mean rate charged to firms in the top quartile by carbon emissions and that charged to firms in the bottom quartile: throughout the sample period, the rates charged to highemission firms exceed those charged to low-emission firms. The difference between the two averages to 14 basis points over the whole period, and ranges from a minimum of 5 to a maximum of 24 basis points. Figure 2 instead compares the rates charged to firms that have not announced an emission target with those charged to firms that have done so (based on the indicator drawn from Refinitiv data): also in this case, the former systematically exceed the latter, the overall difference averaging to 20 basis points, with a minimum of 13 and a maximum of 26 basis points. Hence, on average banks appear to differentiate their lending rates also based on their clients' prospective carbon emissions, not just their current ones – indeed even more so, as the average difference between the rates charged to firms with an emission target and those without a target exceeds that between high-emission and low-emission firms by 6 basis points.

While this preliminary evidence suggests that euro-area banks do price climate risk, it may be vitiated by composition effects, as firms with different carbon footprints may differ in many other respects, such as credit risk, size, location, etc. To take these important concerns into account, in what follows we provide evidence based on panel estimation, which enables us to control for unobserved heterogeneity in firm and bank characteristics. To investigate whether banks price climate risk in their lending rates, in Table 2 we estimate variants of the following specification:

$$r_{f,b,t} = \beta_1 P D_{f,b,t} + \beta_2 Carbon_{f,t} + \beta_3 Target_{f,t} + \beta_4 Carbon_{f,t} \times Target_{f,t} + \theta_{f,b,t} + \epsilon_{f,b,t}, \quad (1)$$

where $r_{f,b,t}$ is the average credit spread charged by bank b on its loans to firm f in month t relative to maturity-matched risk-free rate in the same month: hence, it varies across banks, firms and time. The coefficients β_1 and β_2 capture the credit risk and carbon risk premia respectively, β_3 measures the carbon risk premium differential for firms announcing a target, β_4 allows this differential to depend on firm f's current emissions, $\theta_{f,b,t}$ is a set of firm, bank and time fixed effects, and $\epsilon_{f,b,t}$ is an error term. Including firm fixed effects, hence only relying on within-firm variation, is particularly demanding in our setting, given the length of our sample period. For this reason, we also consider specifications where instead of firm fixed effects we include industry-location-size (ILS) fixed effects, which compare firms with different emissions within the same industrial sector, country and size class. We also present specifications where ILS fixed effects are interacted with time effects as in Degryse, De Jonghe, Jakovljević, Mulier, and Schepens (2019) and Acharya, Eisert, Eufinger, and Hirsch (2019), but cannot include firm fixed effects interacted with time effects as this would prevent identification of the coefficients of interest. Standard errors are clustered at bank-month level.

The estimates reported in Column 1, which includes bank fixed effects and time effects, shows that β_1 and β_2 are positive, while β_3 is negative: banks charge higher credit spreads to firms with greater credit risk, as measured by their average PD, and to firms with higher current carbon emissions, but lower spreads to firms that disclose an emission target, suggesting that they assess the climate risk of their clients not only on the basis of their current abut also their future policies. All coefficient estimates are statistically significant. Column 2 shows that these results also hold upon including ILS fixed effects.

Column 3 shows the estimates of the full specification (1), including the interaction $Carbon_{f,t} \times Target_{f,t}$, to investigate whether the mitigating effect of the firm's commitment depends on its current emissions level: the coefficient β_4 of this interaction is estimated to be negative and strongly significant, indicating that the mitigating effect of a firm's commitment on interest premia is particularly strong if currently the firm is highly polluting. This suggests that bank interest rates are set so as to encourage firms' investments in emission abatement of emissions where such investments are particularly needed.

In the specifications of Columns 4 and 5, we include ILS-time fixed effects instead of ILS fixed effects, and in the last two columns we only exploit within-firm variation by including firm fixed effects. The signs and statistical significance of the estimates remain largely unaffected.

A possible concern about these results is that they may depend on the choice of clustering of the standard errors. However, in Table A1, we cluster standard errors at firm-time level (rather than bank-time level), and results do not change. Another concern may be that specification (1) includes the probability of default $(PD_{f,b,t})$ reported by banks to control for credit risk, and this variable may already capture climate risk, insofar as it may contribute to trigger defaults. But internal risk models are unlikely to take climate risk into account, as they focus on factors that may trigger default within a year, while climate risk tends to refer to rare aggregate events (such as carbon repricing) that typically materialize over a longer horizon. Moreover, banks have no incentives to incorporate climate risk into their internal models, as this may entail additional capital charges not yet required by regulators. Indeed, in our sample, $PD_{f,b,t}$ turns out to be uncorrelated with measures of climate risk: its correlation with $Carbon_{f,t}$ is 0.007 and that with $Target_{f,t}$ is -0.014. In addition, our results are robust to replacing $PD_{f,b,t}$ with a set of firm-level observable characteristics (liquidity, leverage, volatility of ROE and size), as shown in Table A2.

To assess the economic significance of these estimates, in the specification of Column 1 that only includes bank and time fixed effects, the premium on firms with high emissions (at the 90th percentile) is 4 basis points, while the discount for firms committed to reduce emissions is 10 basis points.⁸ This indicates that, in pricing climate risk, banks place greater weight on firms' planned investment to reduce emissions than on their current level of emissions, in line with the descriptive evidence shown in Figures 1 and 2. To place these estimates in perspective, recall that in our sample the standard deviation of the spread is 76 basis points, and notice that the magnitude of the implied climate premia exceeds that of credit risk premia that banks charge to the same firms, the premium for high-PD firms (at the 90th percentile) being only 3 basis points. In the specification shown in Column 3, which includes the ILS fixed effect and controls for the interaction among current emissions and firms' commitment, the premium on emissions is 2 basis points and the discount on the disclosure of a commitment target is 7 basis points, while the premium charged to high PD firms (at the 90th percentile) amounts to 2 basis points.

⁸The order of magnitude of this result is similar to that of the premium for emissions found for syndicated loans to international firms in Ehlers, Packer, and De Greiff (2022). The authors report a magnitude of 3-4 basis points on average and 7 basis points for firms at the 90th percentile of emissions. However, in their international sample the emissions produced by firms are much larger: the 90th percentile corresponds to firms producing more than 1000 tonnes of emissions per million dollars of revenues, while in our European sample firms at the 90th percentile produce 530 tonnes of emissions per million dollars of revenues. For firms producing more than 1000 tonnes per million dollars of revenues, our estimates would also imply a premium of 7 basis points. Magnitudes are similar for the climate risk premia reported for corporate bonds by Huynh and Xia (2021), although they rely on a very different approach: they construct a climate change news beta that captures a bond's covariance with a climate change news risk index and show that bonds with a 1-standard-deviation higher beta feature a 6 basis points reduction in the subsequent month's bond excess return.

While our focus is on the pricing of climate risk, in Table A5 of the Appendix we check what happens if the dependent variable in the panel regressions is the loan amount: using the most saturated specification with firm, bank and time fixed effects, we find that firms with larger emissions receive smaller loans, and the coefficient of the dummy for the firms that announce an emission target is positive but not significant (Columns 1, 2 and 3).

4.2 Role of banks' commitment in climate risk pricing

The next question of interest is whether banks committed to environmental objectives place a higher price on climate risk than non-committed banks. To answer it, we augment specification (1) with additional interactions of banks' SBTi commitment dummy variable $(Commit_{b,t})$ with their customers' PD, current emissions and target emission dummy:

$$r_{f,b,t} = \beta_1 P D_{f,b,t} + \beta_2 Carbon_{f,t} + \beta_3 Target_{f,t} + \beta_4 Carbon_{f,t} \times Target_{f,t}$$

$$+ (\gamma_0 + \gamma_1 P D_{f,b,t} + \gamma_2 Carbon_{f,t} + \gamma_3 Target_{f,t}) \times Commit_{b,t} + \theta_{f,b,t} + \epsilon_{f,b,t},$$

$$(2)$$

In Table 3, which reports the estimates of equation (2), the model is gradually saturated with fixed effects, as in Table 2. Starting from a basic specification with bank and time fixed effects (Column 1), we first add ILS fixed effects (Column 2), then replace them with ILS-time effects (Column 3), next we exploit within-firm variation by including firm fixed effects (Column 4), culminating with the most saturated specification that includes firm-time effects (Column 6).⁹

The estimates indicate that banks committed to decarbonization place a higher price on climate risk than uncommitted banks: they charge a higher premium to polluters, the coefficient γ_2 being positive in all the specifications, and significantly different from zero in those that do not include firm or firm-time effects, which are very demanding given the short time span of our data. Moreover, in all six specifications of the table the coefficient γ_3 is estimated to be negative and statistically significant: committed banks

⁹Notice that here, differently from Table 2, we can include firm fixed effects interacted with time effects because our variables of interest vary at bank level.

charge a lower premium to firms that disclose a target reduction plan. On the other hand, the estimate of the coefficient γ_1 , which refers to the pricing of credit risk by committed banks, varies across specifications and is negative and statistically significant in three of them. This alleviates the concern that the greater premium that banks place on climate risk may simply reflect higher risk aversion: if anything, banks that are more sensitive to climate risk appear to price credit risk less than others.

Banks' environmental commitment also has sizeable quantitative significance for the pricing of climate risk: in the specification of Column 2, which includes bank, time and ILS effects, committed banks reduce the spread by 16 basis points (21% of its standard deviation) more to firms that set an emission target and charge 2 basis points (3% of the premium's standard deviation) more to high emitters, and 1 basis point less than other banks to firms with higher credit risk (i.e., firms whose PD is at the 90th percentile).

Results on quantities are consistent with these results on prices. The panel estimates reported in Table A5, Columns 3 and 4, of the Appendix indicate that committed banks offer lower loan volumes to firms with high emissions and higher volumes to firms with an emission target. These findings differ from those reported by Sastri, Verner, and Marques-Ibanez (2024), who find that banks joining the Net Zero Banking Alliance (NZBA) do not change their volume of lending differently from banks without climate commitments. However, in their paper the NZBA banks do charge 5 bp more to the mining sector, in line with our findings. The differences with our results may be due to the fact that their estimates refer to lending by committed banks at the sector level, while ours refer to interest rates and lending at firm level, distinguishing between current and future emissions, and that the two studies employ a different measure of bank commitment, ours being based on the SBTi agreement.

5 Does Monetary Policy Affect Banks' Pricing of Climate Risk?

The evidence in Section 4 indicates that euro-area banks do price the climate risk of their loans, especially when they publicly commit to decarbonization. In this section we investigate whether the price that they place on climate risk is affected by monetary policy shocks, namely, whether these shocks affect the credit premia they charge to firms with different current and expected emissions and, relatedly, the amount of lending extended to these firms. We start by estimating the contemporaneous response of credit premia to monetary policy shocks via panel regressions (Section 5.1), relying on specifications similar to those used in the previous section. Next, to take into account that credit premia and loan volumes may gradually respond to monetary policy shocks over time, we estimate local projection regressions for both premia and lending volumes (Section 5.2). Taking into account such a delayed response appears warranted for at least two reasons. First, insofar as firms are funded via long-term and medium-term loans, their rates and debt levels are insulated from high-frequency changes in banks' lending policies. Second, monetary policy shocks can be expected to affect the demand for loans with a considerable lag, as firms adjust their production, investment and hiring decisions to take changes in bank lending standards into account (Friedman, 1961).

5.1 Impact Effects

We start by investigating the immediate response of credit spreads to monetary policy surprises. To this purpose, Column 1 of Table 4 reports the estimates of a credit spread regression on monetary policy shocks that includes bank and firm fixed effects, but not time effects, as these would absorb changes in the monetary policy stance (as in fact is the case in other columns of this table).

Since the monetary policy shock, MP_t , is defined as an unexpected increase in policy rates, the positive estimate of its coefficient in Column 1 indicates that contractionary monetary policy shocks are associated with larger credit spreads, as expected. The positive and significant coefficient of the interaction between credit risk and monetary policy surprises, $MP_t \times PD_{f,t}$, indicates that the "traditional" risk-taking channel of monetary policy is indeed at play in our data: restrictive monetary policy leads banks to tighten lending standards more for firms featuring higher credit risk; symmetrically, expansionary monetary policy induce banks to relax their lending standards more for riskier firms. In terms of economic significance, the estimated coefficient implies that a monetary policy surprise of 25 basis points is associated with an increase of 35 basis points in banks' credit spreads.

In subsequent columns of the table, we estimate a richer model, which enables us to investigate the effect of monetary policy shocks on banks' climate risk pricing but include time effects that absorb the direct effect of monetary policy on credit spreads:

$$r_{f,b,t} = \beta_1 P D_{f,b,t} + \beta_2 Carbon_{f,t} + \beta_3 Target_{f,t} + \beta_4 Carbon_{f,t} \times Target_{f,t} + (\delta_1 P D_{f,b,t} + \delta_2 Carbon_{f,t} + \delta_3 Target_{f,t}) \times MP_t + \theta_{f,b,t} + \epsilon_{f,b,t}.$$
(3)

As in previous tables, the specifications are gradually saturated with fixed effects. The estimate of the coefficient δ_2 is positive and significant and that of δ_3 is negative and significant (except in Column 4), which indicates that the risk-taking channel of monetary policy applies not only to credit risk but to climate risk as well: restrictive monetary policy induces banks to place a higher price on climate risk, tightening lending standards more for clients with higher emissions, and less for those that commit to reduce emissions. In the specification with bank, time and ILS fixed effects (Column 3), the increase in bank lending premia associated with a 25 basis-point unexpected rise in policy rates is estimated to be 1.4 basis points larger for high emitters, but 5 basis points smaller for firms committed to decarbonization. By the same token, expansionary monetary policy induces banks to place a lower price on climate risk, relaxing lending standards more for more polluting firms, and less for those committed to reduce emissions.

We find very similar results when monetary policy shocks are measured with dummy

variables capturing the large change in the ECB's stance between the end of 2021 and 2022, instead of relying on high-frequency monetary policy surprises. In Tables A3 and A4 of the Appendix, we estimate the response of banks' lending policies to the quantitative tightening announcement in December 2021, when the ECB announced the end of net security purchases under Pandemic Emergency Purchase Programme (PEPP) and the reduction of the Asset Purchase Programme (APP) net purchases, which may have signaled future policy rate normalisation, and the ECB July 2022 announcement of its first interest rate hike after several years.

The Appendix also reports panel estimates of bank loan volume responses to highfrequency monetary policy surprises. Column 5 of Table A5 shows that tighter monetary policy is associated with a drop of lending volume to all firms, while Column 6 indicates that on impact it does not have a different effect on firms with different exposure to climate risk.

One may wonder whether our findings may to some extent reflect the implementation of the ECB 2022 Climate Stress. All of the eurozone significant banks, which are under the direct supervision of the ECB, were to some extent involved in this exercise, through a qualitative questionnaire aiming at assessing banks' climate risk stress-testing capabilities and through peer benchmarks on banks' exposure to transition risk and to carbon emission-intensive industries. However, only 41 banks actually participated in the stress-test exercise, providing projections, different scenarios and risk areas.¹⁰ In Table A6 we exclude these large 41 banks and find that, even in the remaining sample, banks charge higher interest rates to high-emission firms than to low-emission ones, and lower rates to those that disclose an emission target than to those that do not; moreover, the differential impact of monetary policy is also present, but only for firms with an emission reduction target.

¹⁰See the report on the 2022 climate risk stress test for details: https://www.bankingsupervision.europa.eu/ecb/pub/pdf/ssm.climate_stress_test_report.20220708~2e3cc0999f.en.pdf).

5.2 Dynamic Responses

To evaluate the dynamic impact of monetary policy on lending spreads and loan volumes, we estimate impulse response functions using local projections (LP) methods (Jordà, 2005), which are in general equivalent to those obtained via vector autoregression (VAR) models (Plagborg-Møller and Wolf, 2021). The LP method is flexible enough to accommodate a panel structure in a very simple and computationally convenient way. In practice, we estimate the following panel model:

$$y_{b,f,t+h} = \lambda_{1h}MP_t + \lambda_{2h}MP_t \times Carbon_{f,t} + \lambda_{3h}MP_t \times Target_{f,t} + \theta_b + \epsilon_{f,b,t+h}, \quad (4)$$

where the outcome variable $y_{b,f,t+h}$ is either the lending spread charged or the (logarithm of the) amount lent by bank b to firm f between month t and month t + h; MP_t is our high-frequency measure of monetary policy shocks; θ_b are bank fixed effects. The model includes interaction effects that capture the differential effects of monetary policy depending on firms' exposure to climate risk.

The three upper charts of Figure 3 illustrate the cumulative response of lending spreads to a change in monetary policy at time t from impact up to month t + 12, while the three lower charts illustrate the the cumulative response of lending volumes to the same shock. The figure is drawn for a 25 basis-point shock. The box plots show the distribution of the coefficients λ_{1h} , λ_{2h} , λ_{3h} for each horizon h. The whiskers are the 95% confidence intervals, while the box shows the 90% interval. Standard errors are clustered at the bank-time level.

The first chart in the top panel of Figure 3 shows that the spread charged by banks on loans reacts significantly to the monetary tightening on impact and increases over time until reaching 39 basis points after 1 year. The second chart illustrates the differential effect of the shock for the spread charged to high-emission companies: for firms in the 90th percentile there is an additional increase in spread of around 2 basis points on impact, which decreases to 1 basis points after 1 year. The third chart shows that for firms committed to lowering emissions the increase in premia is reduced by around 5 basis points on impact and about 9 basis points after 1 year.¹¹ Hence, the dynamic effects of the monetary tightening are qualitatively consistent with the estimates reported in Table 4 above, and show that the mitigation of the increase in lending spreads for firms committed to lowering emissions is persistent and increasing over the year after the shock.

The three charts in the bottom panel of the figure show that for loan volumes the effects are negligible on impact, but become statistically and economically significant over time. The estimates in the first chart imply that the effect of the monetary tightening on lending volume reaches 2.5% of the initial lending. The next two charts show that this gradual negative effect is larger for high-emission firms, but smaller for firms that commit to an emission target, for which the interaction term is positive and significant in the last two quarters after the shock. In terms of economic significance, the estimates shown in these two charts imply that a 25-basis-points surprise monetary tightening triggers an additional 2.7% drop in lending for high emitters (those in the top decile by emissions) in a year's time, with a 1.5% mitigation effect for firms that commit to an emission target.¹² Hence, the delayed effects of monetary policy shocks on lending are symmetric in sign with respect to their effects on credit premia shown in the top panel of the figure and in Table 3.

5.3 Survey-based Evidence

Survey-based evidence provides an interesting complementary source of information on the impact of climate change on bank lending to firms in the euro area. Euro-area

¹¹The effects on loan premia after 1 year are computed by multiplying by 25 (basis points) the relevant coefficients and interacted variables. In the first chart, the relevant product is $0.0155 \times 25 = 0.39$; in the second chart, it is $0.0009 \times 0.53 \times 25 = 0.01$, where 0.0009 is the relevant coefficient and 0.53 is the value of emissions for top decile firms; in the third chart, it is $0.0034 \times 1 \times 25 = 0.09$, where 0.0034 is the relevant coefficient.

¹²The percentage effects on volumes after 1 year are computed by multiplying by 25 (basis points) the relevant coefficients and interacted variables. In the first chart, the relevant product is $-0.001 \times 25 = -0.025$; in the second chart, it is $-0.002 \times 0.53 \times 25 = -0.027$, where -0.002 is the relevant coefficient and 0.53 is the value of emissions for top decile firms; in the third chart, it is $0.0006 \times 1 \times 25 = 0.015$, where -0.0006 is the relevant coefficient.

banks provided such self-reported information in responding to the July 2023 round of the Bank Lending Survey (BLS), in which they were asked about the impact of climate change on their lending policies, on top of the standard questions regarding changes in their lending policies. July 2023 is a particularly interesting date for our purposes, since it comes after a whole year of increasingly restrictive monetary policy stance by the ECB, and therefore it may be informative whether euro-area banks perceived their own lending policies as responding differently to the monetary policy shock depending on the different environmental impact of their borrowers.

The BLS is a quarterly survey maintained at the European Central Banks where euro-area banks (158 banks in the July 2023 round) report on the evolution of their internal guidelines or loan approval criteria ("credit standards") and the actual terms and conditions agreed in their loan contracts ("terms and conditions"). Banks' credit standards can be broadly taken to measure their stance in setting loan volumes, while terms and conditions gauge their stance in setting the interest rates charged to their clients. The survey's results are reported in terms of the net percentage of banks changing their lending policies, where net percentage is defined as the difference between the percentage of banks reporting a tightening (an increase) and the percentage of banks reporting an easing (a decrease).

In the July 2023 round of the BLS, conducted between 19 June and 4 July 2023, banks were also required to classify their clients in three groups: "green firms", defined as those that do not contribute or contribute little to climate change; "firms in transition", namely, those that contribute to climate change but are making considerable progress in the transition; and "brown firms", namely, those that contribute significantly to climate change and have not yet started the transition or have made little progress. Then they were asked to indicate how they had changed (if at all) their credit standards and their terms and conditions to each of these three groups of clients over the previous 12 months.

To compare our estimates with the survey results, we match individual responses from euro-area credit institutions surveyed in the BLS to the banks included in our sample. The questionnaire responses indicate that on the whole banks tightened their credit standards as well as terms and conditions for loans or credit lines to firms between July 2022 and July 2023, as one would expect at a time of monetary tightening. More interestingly, the survey results indicate that the tightening of banks' lending policies appears to differentiate between "green firms", "firms in transition" and "brown firms" in a way that is very consistent with the estimates reported in Sections 4 and 5, for interest rates as well as for loan quantities.

First, as shown in top left panel of Figure 4, firms' climate risk appears to have had a net easing impact on their terms and conditions for loans to green firms and, to a lesser extent, for loans to firms in transition, while it had a net tightening impact for loans to brown firms.

Second, as shown in top right panel of the figure, banks committed to decarbonization amplified these changes: committed banks eased more their terms and conditions for loans to green firms and for firms in transition than non-committed banks, and symmetrically tightened them more for brown firms.

Third, the results regarding the impact of climate change on banks' credit standards shown in the two lower panels of the figure are very similar. This indicates that banks differentiated not only their interest rate policies but also their decisions on loan quantities across firms depending on their perceived environmental impact, and that committed banks were more generous in extending credit to green firms and firms in transition than non-committed banks.

6 Conclusion

In this paper we combine euro-area credit register and carbon emission data, to explore (i) whether and to what extent bank interest rates price the climate risk of their client firms, (ii) whether banks committed to decarbonization apply higher prices to the climate risk of their clients, (iii) whether monetary policy shocks impact banks' pricing of climate risk and, if so, in which direction.

We find that euro-area banks charge higher interest rates to firms with larger carbon emissions, and lower rates to firms that commit to green transition, even after controlling for firms' credit risk as measured by their probability of default.

In contrast with other recent findings, banks appear to live up to their word on the issue of climate risk pricing: those that signed a commitment letter within the Science Based Targets initiative (SBTi) indeed provide cheaper loans to firms that commit to decarbonization and, to a smaller extent, penalize more polluting firms.

Finally, we find that the monetary policy of the ECB affects lending to firms not only via a credit risk-taking channel but also via a climate risk-taking channel. Contractionary monetary shocks induce banks to increase both credit risk premia and carbon emission premia, and reduce lending to high emission firms more than to low emission ones. While restrictive monetary policy increases the cost of credit and reduces lending to all firms, its contractionary effect is milder for firms with low emissions and those that commit to decarbonization. These results align quite strikingly with euro-area banks' self-reported information from a survey conducted in July 2023, which shows that banks – especially those committed to decarbonization – differentiate their terms and conditions and their credit standards depending on their clients' perceived environmental impact.

References

- Acharya, Viral, and Hassan Naqvi, 2012, The Seeds of a Crisis: A Theory of Bank Liquidity and Risk Taking over the Business Cycle, *Journal of Financial Economics* 106, 349–366.
- Acharya, Viral V, Tim Eisert, Christian Eufinger, and Christian Hirsch, 2019, Whatever It Takes: The Real Effects of Unconventional Monetary Policy, *The Review of Financial Studies* 32, 3366–3411.
- Adrian, Tobias, and Hyun Song Shin, 2009, Money, liquidity, and monetary policy, American Economic Review 99, 600–605.
- Alogoskoufis, Spyros, Nepomuk Dunz, Tina Emambakhsh, Tristan Hennig, Michiel Kaijser, Charalampos Kouratzoglou, Manuel A Muñoz, Laura Parisi, and Carmelo Salleo, 2021, ECB economy-wide climate stress test: Methodology and results . No. 281 (ECB Occasional Paper).
- Altavilla, Carlo, Luca Brugnolini, Refet S Gürkaynak, Roberto Motto, and Giuseppe Ragusa, 2019, Measuring euro area monetary policy, *Journal of Monetary Economics* 108, 162–179.
- Anderson, Gareth, and Ambrogio Cesa-Bianchi, 2023, Crossing the Credit Channel: Credit Spreads and Firm Heterogeneity, IMF Working Paper 2020/267.
- Baker, Malcolm, Daniel Bergstresser, George Serafeim, and Jeffrey Wurgler, 2022, The Pricing and Ownership of US Green Bonds, Annual Review of Financial Economics 14, 415–437.
- Benincasa, Emanuela, Gazi Kabas, and Steven R. G. Ongena, 2022, 'There is No Planet B', but for Banks 'There are Countries B to Z': Domestic Climate Policy and Cross-Border Lending, Swiss Finance Institute Research Paper Series 22-28.

- Bernanke, Ben, and Mark Gertler, 1989, Agency Costs, Net Worth, and Business Fluctuations, *The American Economic Review* 79, 14–31.
- Bernanke, Ben S., and Mark Gertler, 1995, Inside the Black Box: The Credit Channel of Monetary Policy Transmission, *Journal of Economic Perspectives* 9, 27–48.
- Beyene, Winta, Kathrin De Greiff, Manthos D. Delis, and Steven R. G. Ongena, 2021, Too-Big-To-Strand? Bond Versus Bank Financing in the Transition to a Low-Carbon Economy, CEPR Discussion Paper Series DP16692.
- Bolton, Patrick, Zachery Halem, and Marcin Kacperczyk, 2022, The Financial Cost of Carbon, *Journal of Applied Corporate Finance* 34, 17–29.
- Bolton, Patrick, and Marcin Kacperczyk, 2021, Do Investors Care about Carbon Risk?, Journal of Financial Economics 142, 517–549.

———, 2022, Global Pricing of Carbon-Transition Risk, Journal of Finance forthcoming.

- Bolton, Patrick, and Marcin T. Kacperczyk, 2023, Firm Commitments, SSRN Working Paper 3840813.
- Borio, Claudio, and Haibin Zhu, 2012, Capital Regulation, Risk-taking and Monetary Policy: A Missing Link in the Transmission Mechanism?, *Journal of Financial Stability* 8, 236–251.
- Carbone, Sante, Margherita Giuzio, Sujit Kapadia, Johannes Sebastian Krämer, Ken Nyholm, and Katia Vozian, 2021, The Low-Carbon Transition, Climate Commitments and Firm Credit Risk, ECB Working Paper 2021/2631.
- Degryse, Hans, Olivier De Jonghe, Sanja Jakovljević, Klaas Mulier, and Glenn Schepens, 2019, Identifying Credit Supply Shocks with Bank-Firm Data: Methods and Applications, Journal of Financial Intermediation 40, 100813.

- Degryse, Hans, Roman Goncharenko, Carola Theunisz, and Tamas Vadasz, 2020, When Green Meets Green, National Bank of Belgium Working Paper Research 392 National Bank of Belgium.
- Dell'Ariccia, Giovanni, Luc Laeven, and Robert Marquez, 2014, Real Interest Rates, Leverage, and Bank Risk-Taking, *Journal of Economic Theory* 149, 65–99 Financial Economics.
- Dell'Ariccia, Giovanni, Luc Laeven, and Gustavo A. Suarez, 2017, Bank Leverage and Monetary Policy's Risk-Taking Channel: Evidence from the United States, *The Journal* of Finance 72, 613–654.
- Döttling, Robin, and Adrian Lam, 2024, Monetary Policy, Carbon Transition Risk, and Firm Valuation, SSRN Working Paper 4582767.
- Ehlers, Torsten, Frank Packer, and Kathrin De Greiff, 2022, The Pricing of Carbon Risk in Syndicated Loans: Which Risks Are Priced and Why?, *Journal of Banking & Finance* 136, 106180.
- Engle, Robert F, Stefano Giglio, Bryan Kelly, Heebum Lee, and Johannes Stroebel, 2020,Hedging Climate Change News, *The Review of Financial Studies* 33, 1184–1216.
- Flammer, Caroline, 2021, Corporate Green Bonds, Journal of Financial Economics 142, 499–516.
- Friedman, Milton, 1961, The Lag in Effect of Monetary Policy, Journal of Political Economy 69, 447–466.
- Gambacorta, Leonardo, 2009, Monetary policy and the risk-taking channel, *BIS Quarterly Review*.
- Giannetti, Mariassunta, Martina Jasova, Maria Loumioti, and Caterina Mendicino, 2023, 'Glossy Green' Banks: The Disconnect Between Environmental Disclosures and Lending Activities, SSRN Working Paper Series 4424081.

- Gurkaynak, Refet S., Brian Sack, and Eric T. Swanson, 2005, Do Actions Speak Louder Than Words? The Response of Asset Prices to Monetary Policy Actions and Statements, *International Journal of Central Banking* pp. 55–93.
- Houston, Joel F., and Hongyu Shan, 2021, Corporate ESG Profiles and Banking Relationships, *The Review of Financial Studies* 35, 3373–3417.
- Hsu, Po-Hsuan, Kai Li, and Chi-Yang Tsou, 2023, The Pollution Premium, *The Journal* of Finance 78, 1343–1392.
- Huynh, Thanh D., and Ying Xia, 2021, Climate Change News Risk and Corporate Bond Returns, Journal of Financial and Quantitative Analysis 56, 1985–2009.
- Ilhan, Emirhan, Zacharias Sautner, and Grigory Vilkov, 2020, Carbon Tail Risk, The Review of Financial Studies 34, 1540–1571.
- Iovino, Luigi, Thorsten Martin, and Julien Sauvagnat, 2023, Corporate Taxation and Carbon Emissions, SSRN Working Paper Series 3880057.
- Ivanov, Ivan, Mathias S. Kruttli, and Sumudu W. Watugala, 2023, Banking on Carbon: Corporate Lending and Cap-and-Trade, SSRN Working Paper Series 3650447.
- Jarociński, Marek, and Peter Karadi, 2020, Deconstructing Monetary Policy Surprises—The Role of Information Shocks, American Economic Journal: Macroeconomics 12, 1–43.
- Javadi, Siamak, and Abdullah Al Masum, 2021, The Impact of Climate Change on the Cost of Bank Loans, *Journal of Corporate Finance* 69, 102019.
- Jiménez, Gabriel, Steven Ongena, José-Luis Peydró, and Jesús Saurina, 2014, Hazardous Times for Monetary Policy: What Do Twenty-Three Million Bank Loans Say About the Effects of Monetary Policy on Credit Risk-Taking?, *Econometrica* 82, 463–505.
- Jordà, Oscar, 2005, Estimation and Inference of Impulse Responses by Local Projections, American Economic Review 95, 161–182.

- Kacperczyk, Marcin T., and Jose-Luis Peydro, 2021, Carbon Emissions and the Bank-Lending Channel, SSRN Working Paper 3915486.
- Laeven, Luc A., and Alexander A. Popov, 2022, Carbon Taxes and the Geography of Fossil Lending, ECB Working Paper Series 2022/2762.
- Levine, Ross Eric, Chen Lin, Zigan Wang, and Wensi Xie, 2018, Bank Liquidity, Credit Supply, and the Environment, NBER Working Paper w24375.
- Nguyen, Duc Duy, Steven Ongena, Shusen Qi, and Vathunyoo Sila, 2022, Climate Change Risk and the Cost of Mortgage Credit, *Review of Finance* 26, 1509–1549.
- Papoutsi, Melina, Monica Piazzesi, and Martin Schneider, 2022, How Unconventional Is Green Monetary Policy?, unpublished manuscript.
- Plagborg-Møller, Mikkel, and Christian K. Wolf, 2021, Local projections and vars estimate the same impulse responses, *Econometrica* 89, 955–980.
- Reghezza, Alessio, Yener Altunbas, David Marques-Ibanez, Costanza Rodriguez d'Acri, and Martina Spaggiari, 2022, Do Banks Fuel Climate Change?, *Journal of Financial Stability* 62, 101049.
- Sastri, Parinitha, Emil Verner, and David Marques-Ibanez, 2024, Business as Usual: Bank Climate Commitments, Lending, and Engagement, ECB Working Paper 2921.



Figure 1: Average interest rates charged to high-emission and low-emission firms



Figure 2: Average interest rates charged to firms non-committed and to those committed to lower carbon emissions


Figure 3: Local projection estimates of the response of lending spreads and volumes to a 25-basis-points restrictive monetary policy shock (rise in MP_t)



Figure 4: Net percentages of banks reporting changes in terms and conditions or credit standards in the past 12 months, based on the July 2023 BLS Survey.

Table 1: Summary statistics

This table shows means, standard deviations and number of observations for the variables used in the empirical analysis. Data are at the monthly frequency, and the sample period is from September 2018 to December 2022. $Spread_{b,f,t}$ is the monthly average credit spread charged by bank b on its loans to firm f in month t relative to maturity-matched risk-free rate in that month. PD_{f,t} is the weighted average probability of default of firm f in month t reported in AnaCredit. $Carbon_{f,t}$ are Scope 1 and Scope 2 CO2 (and CO2 equivalents) emissions of firm f in month t in thousands of tonnes divided by the firm's net sales in million US dollars in the same month. $Target_{f,t}$ is a dummy variable that equals 1 if firm f has disclosed a target emission reduction, and 0 otherwise. $Liquidity_{f,t}$ is the firm liquidity ratio calculated as (Current assets - Stocks) / Current liabilities. $Capital_{f,t}$ is the firm capital ratio calculated as Shareholder funds / Total assets. $\sigma(ROE)_{f,t}$ is defined as the standard deviation of the ROE over the previous two years. $Log(Assets)_{f,t}$ is the logarithm of the firm's Total Assets. $Commit_{b,t}$ is a dummy variable that equals 1 if bank b has committed to decarbonization within the Science Based Targets initiative (SBTi) as of month t, and 0 otherwise. MP_t is a monthly measure of the monetary policy surprises extracted from high-frequency intraday yields at short-term maturity at dates of policy announcements in month t. The sample is restricted to publicly listed firms for which data for CO2 emissions and the emission target disclosure are available in the Refinitiv database.

| Variables | Observations | Mean | St. Dev. | p5 | p10 | p25 | p50 | p75 | p90 | p95 |
|-----------------------|--------------|-------|----------|-------|-------|-------|-------|-------|-----------|-----------|
| $Spread_{b,f,t}$ | 325,180 | 1.51 | 0.76 | 0.18 | 0.54 | 1.08 | 1.55 | 2.00 | 2.41 | 2.76 |
| $Log(Credit)_{b,f,t}$ | 398,736 | 0.98 | 2.44 | -3.07 | -1.50 | -0.02 | 1.07 | 2.56 | 3.87 | 4.54 |
| $PD_{f,t}$ | $442,\!469$ | 0.96 | 3.49 | 0.07 | 0.09 | 0.15 | 0.26 | 0.50 | 1.18 | 2.48 |
| $Carbon_{f,t}$ | $435,\!263$ | 0.18 | 0.47 | 0.00 | 0.00 | 0.01 | 0.03 | 0.09 | 0.53 | 0.82 |
| $Target_{f,t}$ | $453,\!231$ | 0.58 | 0.49 | 0.00 | 0.00 | 0.00 | 1.00 | 1.00 | 1.00 | 1.00 |
| $Liquidity_{f,t}$ | $443,\!476$ | 1.85 | 2.36 | 0.31 | 0.40 | 0.68 | 1.06 | 1.87 | 4.09 | 6.08 |
| $Capital_{f,t}$ | $447,\!145$ | 41.42 | 16.45 | 15.47 | 20.46 | 29.32 | 41.47 | 51.31 | 62.42 | 71.81 |
| $\sigma(ROE)_{f,t}$ | $433,\!995$ | 5.08 | 5.75 | 0.00 | 0.15 | 0.97 | 3.19 | 7.29 | 12.32 | 16.45 |
| $Log(Assets)_{f,t}$ | $447,\!145$ | 22.21 | 1.49 | 19.91 | 20.47 | 21.16 | 22.06 | 23.06 | $24,\!27$ | $25,\!12$ |
| $Commit_{b,t}$ | $453,\!231$ | 0.11 | 0.31 | 0.00 | 0.00 | 0.00 | 0.00 | 0.00 | 1.00 | 1.00 |
| MP_t (basis points) | $453,\!231$ | 1.09 | 5.56 | -1.53 | -1.20 | -0.53 | 0.00 | 0.06 | 4.21 | 14.14 |

Table 2: Do Banks Price Climate Risk?

This table reports estimates of regressions of the monthly average credit spread charged by bank b on loans to firm f in month t (Spread_{b,f,t}) on measures of credit and climate risk of its client firms. $PD_{f,t}$, $Carbon_{f,t}$ and $Target_{f,t}$ are defined as in Table 1. ILS Fixed Effects stands for the interaction of the firm's industry (2-digit NACE code), country and size effects. Standard errors clustered at the bank-time level are shown in parentheses below the respective coefficient estimate. *** p<0.01, ** p<0.05, * p<10.

| | (1) | (2) | (3) | (4) | (5) | (9) | (2) |
|-----------------------------------|----------------------------|----------------------------|-----------------|-----------------|----------------|----------------------------|----------------------------|
| $PD_{f,t}$ | 0.0241^{***} | 0.0167^{***} | 0.0167^{***} | 0.0260^{***} | 0.0261^{***} | 0.00528^{***} | 0.00527^{***} |
| × 2 | (0.000543) | (0.000592) | (0.000592) | (0.000770) | (0.000770) | (0.000641) | (0.000641) |
| $Carbon_{f,t}$ | 0.0709^{***} | 0.0199^{***} | 0.0432^{***} | 0.0187^{***} | 0.0898^{***} | 0.0334^{**} | 0.0861^{***} |
| , S | (0.00261) | (0.00615) | (0.00884) | (0.00664) | (0.0118) | (0.0169) | (0.0201) |
| $Target_{f,t}$ | -0.103^{***} | -0.0667*** | -0.0681^{***} | -0.0671^{***} | -0.0778*** | -0.0345^{***} | -0.0342^{***} |
| × 5 | (0.00244) | (0.00254) | (0.00260) | (0.00281) | (0.00323) | (0.00340) | (0.00339) |
| $Carbon_{f,t} 	imes Target_{f,t}$ | | | -0.0319^{***} | | -0.103^{***} | | -0.0454^{***} |
| - - | | | (0.00804) | | (0.0139) | | (0.00860) |
| Bank Fixed Effects | Yes | Yes | Yes | Yes | Yes | Yes | $\mathbf{Y}_{\mathbf{es}}$ |
| Time Fixed Effects | $\mathbf{Y}_{\mathbf{es}}$ | $\mathbf{Y}_{\mathbf{es}}$ | \mathbf{Yes} | I | I | $\mathbf{Y}_{\mathbf{es}}$ | ${ m Yes}$ |
| ILS Fixed Effects | I | Y_{es} | \mathbf{Yes} | I | I | I | ı |
| ILS ×Time Fixed Effects | I | ı | I | \mathbf{Yes} | \mathbf{Yes} | I | ı |
| Firm Fixed Effects | I | ı | I | I | I | $\mathbf{Y}_{\mathbf{es}}$ | ${ m Yes}$ |
| Observations | 306871 | 306788 | 306788 | 305401 | 305401 | 306864 | 306864 |
| R^{2} | 0.468 | 0.550 | 0.550 | 0.602 | 0.603 | 0.617 | 0.617 |
| | | | | | | | |

Table 3: Do Committed Banks Place a Higher Price on Climate Risk?

This table reports estimates of regressions of the monthly average credit spread charged by bank b on loans to firm f in month t (Spread_{b,f,t}) on measures of credit and climate risk of its client firms and of the bank's commitment to reducing carbon emissions. $PD_{f,t}$, $Carbon_{f,t}$, $Target_{f,t}$ and $Commit_{b,t}$ are defined as in Table 1. ILS Fixed Effects stands for the interaction of the firm's industry (2-digit NACE code), country and size effects. Standard errors clustered at the bank-time level are shown in parentheses below the respective coefficient estimate. *** p<0.01, ** p<0.05, * p<0.10.

| | (1) | (2) | (3) | (4) | (5) |
|--|-----------------|------------------|----------------------------|-----------------|-----------------|
| $PD_{f,t}$ | 0.0248^{***} | 0.0176^{***} | 0.0270^{***} | 0.00512^{***} | |
| `````````````````````````````````````` | (0.000566) | (0.000627) | (0.000794) | (0.000660) | |
| $Carbon_{f,t}$ | 0.0414^{***} | 0.0313^{***} | 0.0815^{***} | 0.0823^{***} | |
| × • | (0.00730) | (0.00907) | (0.0121) | (0.0200) | |
| $Target_{f,t}$ | -0.0913^{***} | -0.0591^{***} | -0.0750^{***} | -0.0238*** | |
| ` | (0.00267) | (0.00267) | (0.00331) | (0.00340) | |
| $Commit_{b,t}$ | 0.241^{***} | 0.207^{***} | 0.0175 | 0.213^{***} | 0.0133 |
| | (0.0247) | (0.0235) | (0.0223) | (0.0234) | (0.0210) |
| $Carbon_{f,t} \times Target_{f,t}$ | -0.0328*** | -0.0229*** | -0.0999*** | -0.0394^{***} | |
| ~ • | (0.00767) | (0.00796) | (0.0139) | (0.00852) | |
| $Commit_{b,t} \times PD_{f,t}$ | -0.00669*** | -0.00744^{***} | -0.00772^{***} | 0.000438 | 0.00500^{***} |
| × • | (0.00174) | (0.00151) | (0.00152) | (0.00149) | (0.00144) |
| $Commit_{b,t} \times Carbon_{f,t}$ | 0.0336^{***} | 0.0339^{***} | 0.0310^{***} | 0.00158 | 0.00907 |
| | (0.0115) | (0.0115) | (0.00936) | (0.0124) | (0.0100) |
| $Commit_{b,t} \times Target_{f,t}$ | -0.166^{***} | -0.157^{***} | -0.0572^{***} | -0.163^{***} | -0.0431^{***} |
| | (0.0194) | (0.0203) | (0.0154) | (0.0205) | (0.0146) |
| Bank Fixed Effects | Yes | Yes | Yes | \mathbf{Yes} | \mathbf{Yes} |
| Time Fixed Effects | ${ m Yes}$ | ${ m Yes}$ | ı | ${ m Yes}$ | I |
| ILS Fixed Effects | ı | ${ m Yes}$ | ı | ı | I |
| $ILS \times Time Effects$ | ı | ı | $\mathbf{Y}_{\mathbf{es}}$ | ı | I |
| Firm Fixed Effects | ı | ı | · | ${ m Yes}$ | I |
| Firm \times Time Effects | I | I | ı | I | \mathbf{Yes} |
| Observations | 306871 | 306788 | 305401 | 306864 | 303466 |
| R-squared | 0.460 | 0.551 | 0 603 | 0.618 | 0.607 |

Table 4: How Does Monetary Policy Impact the Pricing of Climate Risk?

This table reports estimates of regressions of the monthly average credit spread charged by bank b on loans to firm f in month t (Spread_{b,f,t}) on measures of credit and climate risk of its client firms, and on a measure of monetary policy surprises. $PD_{f,t}$, $Carbon_{f,t}$, $Target_{f,t}$ and MP_t are defined as in Table 1. ILS Fixed Effects stands for the interaction of the firm's industry (2-digit NACE code), country and size effects. Standard errors clustered at the bank-time level are shown in parentheses below the respective coefficient estimate. *** p<0.01, ** p<0.05, * p<10.

| | (1) | (2) | (3) | (4) | (5) |
|--|-----------------|------------------|------------------|-----------------|------------------|
| $PD_{f,t}$ | 0.00777^{***} | 0.0242^{***} | 0.0168^{***} | 0.0261^{***} | 0.00540^{***} |
| × • | (0.000724) | (0.000546) | (0.000593) | (0.000769) | (0.000643) |
| $Carbon_{f,t}$ | | 0.0506^{***} | 0.0425^{***} | 0.0893^{***} | 0.0856^{***} |
| × > | | (0.00758) | (0.00885) | (0.0118) | (0.0201) |
| $Target_{f,t}$ | | -0.103^{***} | -0.0688*** | -0.0780*** | -0.0349^{***} |
| ` | | (0.00252) | (0.00260) | (0.00323) | (0.00340) |
| $Carbon_{f,t} 	imes Target_{f,t}$ | | -0.0260^{***} | -0.0308^{***} | -0.102^{***} | -0.0443^{***} |
| | | (0.00788) | (0.00806) | (0.0139) | (0.00862) |
| MP_t | 0.0150^{***} | ~ | | ~ | ~ |
| | (0.000876) | | | | |
| $MP_t 	imes PD_{f,t}$ | 0.000263^{**} | 0.000399^{***} | 0.000348^{***} | 0.000340^{**} | 0.000274^{***} |
| `````````````````````````````````````` | (0.000118) | (0.000110) | (0.000105) | (0.000154) | (0.0000914) |
| $MP_t 	imes Carbon_{f,t}$ | | 0.00111^{*} | 0.00107^{*} | 0.00233^{*} | 0.000990^{*} |
| × 3 | | (0.000673) | (0.000587) | (0.00138) | (0.000585) |
| $MP_t 	imes Target_{f,t}$ | | -0.00329*** | -0.00205^{***} | -0.000509 | -0.00162^{***} |
| 2 | | (0.000575) | (0.000554) | (0.000686) | (0.000528) |
| Bank Fixed Effects | Yes | Yes | Yes | \mathbf{Yes} | Yes |
| Time Fixed Effects | I | Y_{es} | ${ m Yes}$ | ı | Yes |
| ILS Fixed Effects | I | ı | \mathbf{Yes} | ı | I |
| $ILS \times Time Fixed Effects$ | ı | ı | ı | ${ m Yes}$ | ı |
| Firm Fixed Effects | \mathbf{Yes} | ı | ı | ı | \mathbf{Yes} |
| Observations | 321331 | 306871 | 306788 | 305401 | 306864 |
| R-squared | 0.366 | 0.468 | 0.550 | 0.603 | 0.617 |

Appendix

Table A1: Do Banks Price Climate Risk? Alternative clustering

This table reports estimates of regressions of the monthly average credit spread charged by bank b on loans to firm f in month t (Spread_{b,f,t}) on measures of credit and climate risk of its client firms. $PD_{f,t}$, $Carbon_{f,t}$ and $Target_{f,t}$ are defined as in Table 1. ILS Fixed Effects stands for the interaction of the firm's industry (2-digit NACE code), country and size effects. Standard errors clustered at the firm-time level are shown in parentheses below the respective coefficient estimate. *** p<0.01, ** p<0.05, * p<10.

| | (1) | (2) | (3) | (4) | (5) | (9) | (2) |
|---|----------------|----------------------------|----------------------------|-----------------|----------------|-----------------|-----------------|
| $PD_{f,t}$ | 0.0241^{***} | 0.0167^{***} | 0.0167^{***} | 0.0260^{***} | 0.0261^{***} | 0.00528^{***} | 0.00527^{***} |
| × • | (0.000928) | (0.00125) | (0.00125) | (0.00155) | (0.00156) | (0.00103) | (0.00103) |
| $Carbon_{f,t}$ | 0.0709^{***} | 0.0199^{*} | 0.0432^{***} | 0.0187 | 0.0898^{***} | 0.0334 | 0.0861^{***} |
| ~ | (0.00736) | (0.0114) | (0.0154) | (0.0115) | (0.0191) | (0.0229) | (0.0268) |
| $Target_{f,t}$ | -0.103^{***} | -0.0667*** | -0.0681^{***} | -0.0671^{***} | -0.0778*** | -0.0345^{***} | -0.0342^{***} |
| 2 | (0.00789) | (0.00804) | (0.00801) | (0.00966) | (0.00969) | (0.00753) | (0.00753) |
| $Carbon_{f,t} \times Target_{f,t}$ | | | -0.0319^{**} | | -0.103^{***} | | -0.0454^{***} |
| | | | (0.0131) | | (0.0232) | | (0.0131) |
| Bank FE | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Firm FE | ı | ı | ı | ı | ı | \mathbf{Yes} | \mathbf{Yes} |
| Time FE | ${ m Yes}$ | $\mathbf{Y}_{\mathbf{es}}$ | $\mathbf{Y}_{\mathbf{es}}$ | \mathbf{Yes} | ${ m Yes}$ | \mathbf{Yes} | \mathbf{Yes} |
| $\operatorname{Firm}\operatorname{ILS}$ | ı | $\mathbf{Y}_{\mathbf{es}}$ | $\mathbf{Y}_{\mathbf{es}}$ | ı | ı | ı | I |
| Firm ILS * Time FE | ı | ı | ı | \mathbf{Yes} | \mathbf{Yes} | ı | I |
| Observations | 306871 | 306788 | 306788 | 305401 | 305401 | 306864 | 306864 |
| R-squared | 0.468 | 0.550 | 0.550 | 0.602 | 0.603 | 0.617 | 0.617 |

Table A2: Do Banks Price Climate Risk? Firm Control Variables Instead of PD

This table reports estimates of regressions of the monthly average credit spread charged by bank b on loans to firm f in month t (Spread_{b, f,t}) on measures of climate risk and other firm observable variables of its client firms. $PD_{f,t}$, $Carbon_{f,t}$, $Target_{f,t}$, $Liquidity_{f,t-1}$, $\sigma(ROE)_{f,t-1}$ and $Log(Assets)_{f,t-1}$ are defined as in Table 1. ILS Fixed Effects stands for the interaction of the firm's industry (2-digit NACE code), country and size effects. Standard errors clustered at the bank-time level are shown in parentheses below the respective coefficient estimate. *** p<0.01, ** p<0.05, ** p<10.

| | (1) | (2) | (3) | (4) | (5) | (9) | (2) |
|-----------------------------------|------------------|------------------|-----------------|------------------|------------------|----------------------------|----------------------------|
| $Carbon_{f,t}$ | 0.0927^{***} | 0.0134^{**} | 0.0243^{***} | 0.0228^{***} | 0.0590^{***} | 0.00446 | 0.0456^{**} |
| × • | (0.00265) | (0.00619) | (0.00876) | (0.00667) | (0.0123) | (0.0168) | (0.0198) |
| $Target_{f,t}$ | -0.0895*** | -0.0734^{***} | -0.0742^{***} | -0.0743^{***} | -0.0796*** | -0.0278*** | -0.0278*** |
| | (0.00253) | (0.00270) | (0.00276) | (0.00307) | (0.00352) | (0.00359) | (0.00359) |
| $Carbon_{f,t} 	imes Target_{f,t}$ | | | -0.0147^{*} | | -0.0515^{***} | | -0.0345^{***} |
| | | | (0.00805) | | (0.0145) | | (0.00860) |
| $Liquidity_{f,t-1}$ | -0.00590*** | -0.00593^{***} | -0.00597*** | -0.0115^{***} | -0.0115^{***} | 0.00640^{***} | 0.00628^{***} |
| × 2 | (0.000508) | (0.000555) | (0.000555) | (0.000685) | (0.000686) | (0.000645) | (0.000644) |
| $Capital_{f,t-1}$ | -0.00312^{***} | -0.00397*** | -0.00397*** | -0.00409^{***} | -0.00408^{***} | -0.00279*** | -0.00278*** |
| د ، ۱ | (0.0000771) | (0.000102) | (0.000103) | (0.000110) | (0.000110) | (0.000171) | (0.000171) |
| $\sigma(ROE)_{f,t-1}$ | 0.000169 | 0.00119^{***} | 0.00122^{***} | 0.000621^{**} | 0.000702^{***} | 0.00307^{***} | 0.00317^{***} |
| × 3 3 | (0.000210) | (0.000220) | (0.000220) | (0.000253) | (0.000253) | (0.000273) | (0.000273) |
| $Log(Assets)_{f,t-1}$ | -0.0648^{***} | -0.0560*** | -0.0562^{***} | -0.0567*** | -0.0572^{***} | -0.0117^{***} | -0.0128^{***} |
| ~ > | (0.00106) | (0.00254) | (0.00254) | (0.00268) | (0.00270) | (0.00444) | (0.00443) |
| Bank Fixed Effects | Yes | Yes | Yes | Yes | Yes | Yes | Yes |
| Time Fixed Effects | \mathbf{Yes} | \mathbf{Yes} | \mathbf{Yes} | ı | I | Yes | \mathbf{Yes} |
| ILS Fixed Effects | ı | \mathbf{Yes} | \mathbf{Yes} | ı | ı | ı | ı |
| ILS \times Time Fixed Effects | ı | ı | · | \mathbf{Yes} | \mathbf{Yes} | ı | ı |
| Firm Fixed Effects | I | I | I | I | I | $\mathbf{Y}_{\mathbf{es}}$ | $\mathbf{Y}_{\mathbf{es}}$ |
| Observations | 291402 | 291399 | 291399 | 289865 | 289865 | 291394 | 291394 |
| R^2 | 0.475 | 0.557 | 0.557 | 0.605 | 0.605 | 0.625 | 0.625 |

| itening of | |
|----------------------------|------|
| enin | |
| | |
| ve Tigl | |
| itati | |
| lanti | |
| e Qu | |
| o th | |
| on t | |
| acti | |
| Re | |
| e Risk? Reaction to the Qu | |
| | _ |
| Climat | 2021 |
| of | ber |
| icing . | em |
| Pric | Ŏ |
| \mathbf{the} | |
| pact | |
| Im] | |
| olicy | |
| y P | |
| onetar | |
| H | |
| Mc | |
| oes M | |
| w Does Mc | |
| oes M | |
| : How Does M | |
| oes M | |

This table reports estimates of regressions of the monthly average credit spread charged by bank b on loans to firm f in month t (Spread, f, f) on measures undertaken under the Pandemic Emergency Purchase Programme (PEPP) and the reduction of Asset Purchase Programme (APP) net purchases, and 0 of credit and climate risk of its client firms and a dummy Post- QT_t which equals 1 since December 2021 when the ECB announced the end of net purchases NACE code), country and size effects. Standard errors clustered at the bank-time level are shown in parentheses below the respective coefficient estimate. *** otherwise. $PD_{f,t}$, $Carbon_{f,t}$, $Target_{f,t}$ and $Commit_{b,t}$ are defined as in Table 1. ILS Fixed Effects stands for the interaction of the firm's industry (2-digit p<0.01, ** p<0.05, * p<10.

| | (1) | (2) | (3) | (4) | (5) |
|-----------------------------------|-----------------|-----------------|---|---|----------------------------|
| $PD_{f,t}$ | 0.00229^{***} | 0.0222^{***} | 0.0150^{***} | 0.0237^{***} | 0.00217^{***} |
| × 2 | (0.000645) | (0.000594) | (0.000625) | (0.000861) | (0.000633) |
| $Carbon_{f,t}$ | | 0.000535 | 0.00674 | 0.140^{***} | 0.0163 |
| 2 | | (0.00787) | (0.0104) | (0.0126) | (0.0215) |
| $Target_{f,t}$ | | -0.0948^{***} | -0.0604^{***} | -0.0837^{***} | -0.0281^{***} |
| × • | | (0.00264) | (0.00276) | (0.00338) | (0.00354) |
| $Carbon_{f,t} 	imes Target_{f,t}$ | | -0.0729*** | -0.00561 | -0.149*** | -0.0238^{**} |
| | | (0.00813) | (0.0101) | (0.0140) | (0.0110) |
| $Post-QT_t$ | 0.840^{***} | | | | |
| | (0.00940) | | | | |
| $Post-QT_t 	imes PD_{f,t}$ | 0.00405^{***} | 0.00604^{***} | 0.00550^{***} | 0.00831^{***} | 0.00890^{***} |
| 2 | (0.00114) | (0.00124) | (0.00119) | (0.00179) | (0.00104) |
| $Post-QT_t 	imes Carbon_{f,t}$ | | 0.0788^{***} | 0.0563^{***} | 0.0889^{***} | 0.101^{***} |
| | | (0.0104) | (0.0108) | (0.0186) | (0.0113) |
| $Post-QT_t 	imes Target_{f,t}$ | | -0.0389^{***} | -0.0501^{***} | -0.00771 | -0.0467^{***} |
| | | (0.00838) | (0.00816) | (0.0103) | (0.00822) |
| Bank Fixed Effects | \mathbf{Yes} | \mathbf{Yes} | \mathbf{Yes} | \mathbf{Yes} | \mathbf{Yes} |
| Time Fixed Effects | I | \mathbf{Yes} | \mathbf{Yes} | I | $\mathbf{Y}_{\mathbf{es}}$ |
| ILS Fixed Effects | I | ı | \mathbf{Yes} | I | I |
| $ILS \times Time Fixed Effects$ | ı | ı | I | \mathbf{Yes} | I |
| Firm Fixed Effects | \mathbf{Yes} | I | I | I | $\mathbf{Y}_{\mathbf{es}}$ |
| Observations | 321331 | 306871 | 306788 | 305401 | 306864 |
| R-squared | 0.523 | 0.468 | 0.550 | 0.603 | 0.617 |
| | >->-> | >>++>> | >>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>> | >>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>> | , , |

Table A4: How Does Monetary Policy Impact the Pricing of Climate Risk? Reaction to the Rate Hike of July 2022

This table reports estimates of regressions of the monthly average credit spread charged by bank b on loans to firm f in month t (Spread_{b,f,t}) on measures of credit and climate risk of its client firms and a dummy $Post-Hike_t$, which equals 1 since July 2022 when the ECB announced the first increase in its policy rate after several years, and 0 otherwise. $PD_{f,t}$, $Carbon_{f,t}$, $Target_{f,t}$ and $Commit_{b,t}$ are defined as in Table 1. ILS Fixed Effects stands for the interaction of the firm's industry (2-digit NACE code), country and size effects. Standard errors clustered at the bank-time level are shown in parentheses below the respective coefficient estimate. *** p<0.01, ** p<0.05, * p<10.

| | (1) | (2) | (3) | (4) | (5) |
|---------------------------------------|----------------------------|-----------------|----------------|-----------------|----------------------------|
| $PD_{f,t}$ | 0.000830 | 0.0235^{***} | 0.0162^{***} | 0.0253^{***} | 0.00389^{***} |
| × \$ | (0.000624) | (0.000540) | (0.000577) | (0.000779) | (0.000591) |
| $Carbon_{f,t}$ | | 0.0203^{***} | 0.0119 | 0.0798^{***} | 0.0503^{**} |
| , , | | (0.00749) | (0.00903) | (0.0123) | (0.0201) |
| $Target_{f,t}$ | | -0.0998*** | -0.0665*** | -0.0784*** | -0.0349^{***} |
| ×. 9 | | (0.00253) | (0.00262) | (0.00327) | (0.00340) |
| $Carbon_{f,t} 	imes Target_{f,t}$ | | -0.0558^{***} | -0.00178 | -0.0934^{***} | -0.000946 |
| ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ | | (0.00781) | (0.00829) | (0.0142) | (0.00886) |
| $Post-Hike_t$ | 0.254^{***} | ~ | ~ | ~ | ~ |
| | (0.0108) | | | | |
| $Post-Hike_t \times PD_{f,t}$ | 0.00928^{***} | 0.00449^{**} | 0.00295 | 0.00768^{**} | 0.00799^{***} |
| × 5 | (0.00190) | (0.00203) | (0.00202) | (0.00323) | (0.00188) |
| $Post-Hike_t \times Carbon_{f,t}$ | | 0.123^{***} | 0.121^{***} | 0.0529 | 0.155^{***} |
| × 2 | | (0.0209) | (0.0194) | (0.0331) | (0.0193) |
| $Post-Hike_t \times Target_{f,t}$ | | -0.0139 | -0.0255 | -0.0447* | -0.0380^{*} |
| | | (0.0221) | (0.0220) | (0.0235) | (0.0219) |
| Bank Fixed Effects | \mathbf{Yes} | Yes | Yes | Yes | \mathbf{Yes} |
| Time Fixed Effects | ı | \mathbf{Yes} | \mathbf{Yes} | I | $\mathbf{Y}_{\mathbf{es}}$ |
| ILS Fixed Effects | I | · | Yes | ı | I |
| $ILS \times Time Fixed Effects$ | ı | ı | ı | \mathbf{Yes} | I |
| Firm Fixed Effects | $\mathbf{Y}_{\mathbf{es}}$ | ı | I | I | \mathbf{Yes} |
| Observations | 321331 | 306871 | 306788 | 305401 | 306864 |
| R-somared | 0.508 | 0.468 | 0.550 | 0.603 | 0.617 |

Table A5: Climate Risk, Committment and Monetary Policy: Effects on Quantities

on measures of credit and climate risk of its client firms and on a measure of monetary policy surprises. $PD_{f,t}$, $Carbon_{f,t}$, $Target_{f,t}$, $Commit_{b,t}$ and MP_t are defined as in Table 1. Standard errors clustered at the bank-time level are shown in parentheses below the respective coefficient estimate. *** p<0.01, ** This table reports estimates of regressions of the monthly average of credit exposure (log of loan amount) by bank b to firm f in month t (Log(Credit) $_{b,f,t}$) p<0.05, * p<10.

| | (1) | (2) | (3) | (4) | (5) | (9) |
|------------------------------------|----------------------------|----------------|----------------------|--------------------------|--------------------------------|----------------------------|
| $PD_{f,t}$ | -0.00110 | -0.00111 | -0.00347^{**} | | 0.000791 | -0.00119 |
| × 5 | (0.00160) | (0.00161) | (0.00158) | | (0.00163) | (0.00161) |
| $Carbon_{f,t}$ | -0.326^{***} | -0.261^{***} | -0.320^{***} | | | -0.260^{***} |
| х Э | (0.0618) | (0.0688) | (0.0694) | | | (0.0688) |
| $Target_{f,t}$ | 0.00263 | 0.00220 | 0.00161 | | | -0.00169 |
| х Э | (0.00975) | (0.00973) | (0.00957) | | | (0.00974) |
| $Carbon_{f,t} 	imes Target_{f,t}$ | | 0.0565^{**} | 0.0355 | | | 0.0571^{**} |
| | | (0.0254) | (0.0256) | | | (0.0256) |
| $Commit_{b,t}$ | | | 0.156^{***} | 0.134^{***} | | |
| $Commit_{1,1} \times PD_{2,1}$ | | | (0.0594)0 0187 $***$ | (0.0637) 0.0183 $***$ | | |
| | | | (0.00469) | (0.00426) | | |
| $Commit_{b,t} 	imes Carbon_{f,t}$ | | | -0.285^{***} | -0.242** | | |
| ` | | | (0.0340) | (0.0333) | | |
| $Commit_{b,t} \times Target_{f,t}$ | | | 0.0113 | 0.0764^{**} | | |
| | | | (0.0406) | (0.0387) | | |
| MP_t | | | | | -0.00195^{***} (0.000516) | |
| $MP_t 	imes PD_{f,t}$ | | | | | -0.000233 | -0.000160 |
| ×. | | | | | (0.000205) | (0.000208) |
| $MP_t 	imes Carbon_{f,t}$ | | | | | | -0.000594 |
| | | | | | | (0.00144) |
| $MP_t 	imes Target_{f,t}$ | | | | | | 0.00146 |
| | | | | | | (0.00142) |
| Bank FE | ${ m Yes}$ | \mathbf{Yes} | \mathbf{Yes} | Yes | \mathbf{Yes} | \mathbf{Yes} |
| Firm FE | ${ m Yes}$ | \mathbf{Yes} | \mathbf{Yes} | I | \mathbf{Yes} | \mathbf{Yes} |
| Time FE | $\mathbf{Y}_{\mathbf{es}}$ | \mathbf{Yes} | \mathbf{Yes} | I | I | $\mathbf{Y}_{\mathbf{es}}$ |
| Firm $*$ Time FE | I | I | I | \mathbf{Yes} | I | I |
| Observations | 306864 | 306864 | 306864 | 303466 | 321331 | 306864 |
| R-squared | 0.544 | 0.544 | 0.544 | 0.610 | 0.543 | 0.544 |
| | | | | | | |

Table A6: Cimate Risk and Monetary Policy Excluding Banks Subject to Climate Stress Test

This table reports estimates of regressions of the monthly average credit spread charged by bank b on loans to firm f in month t (Spread_{b,f,t}) on measures of credit and climate risk of its client firms, and on a measure of monetary policy surprises. $PD_{f,t}$, $Carbon_{f,t}$, $Target_{f,t}$, $Commit_{b,t}$ and MP_{t} are defined as in NACE code), country and size effects. Standard errors clustered at the bank-time level are shown in parentheses below the respective coefficient estimate. *** Table 1. The sample excludes 41 banks subject to the ECB 2022 Climate Stress Tests. ILS Fixed Effects stands for the interaction of the firm's industry (2-digit p<0.01, ** p<0.05, * p<10.

| | (1) | (2) | (3) | (4) |
|-----------------------------------|-----------------|----------------------------|----------------------------|------------------|
| $PD_{f,t}$ | 0.0299^{***} | 0.00459^{***} | 0.0300^{***} | 0.00475^{***} |
| | (966000.0) | (0.000693) | (0.000995) | (0.000695) |
| $Carbon_{f,t}$ | 0.00850 | 0.186^{***} | 0.00838 | 0.186^{***} |
| ŝ | (0.0226) | (0.0309) | (0.0227) | (0.0309) |
| $Target_{f,t}$ | -0.0981^{***} | -0.0260^{***} | -0.0986*** | -0.0267^{***} |
| × • | (0.00443) | (0.00338) | (0.00445) | (0.00338) |
| $Carbon_{f,t} 	imes Target_{f,t}$ | -0.129^{***} | -0.0857*** | -0.129^{***} | -0.0858*** |
| | (0.0249) | (0.00972) | (0.0250) | (0.00974) |
| $MP_t	imes PD_{f,t}$ | | | 0.000473^{**} | 0.000309^{***} |
| | | | (0.000191) | (0.000101) |
| $MP_t 	imes Carbon_{f,t}$ | | | 0.00274 | -0.0000171 |
| 3 | | | (0.00248) | (0.000516) |
| $MP_t 	imes Target_{f,t}$ | | | -0.00164^{**} | -0.00127^{**} |
| × • | | | (0.000701) | (0.000519) |
| Bank Fixed Effects | Yes | Yes | Yes | Yes |
| Time Fixed Effects | I | \mathbf{Yes} | ı | \mathbf{Yes} |
| $ILS \times Time Fixed Effects$ | \mathbf{Yes} | ı | $\mathbf{Y}_{\mathbf{es}}$ | ı |
| Firm Fixed Effects | I | $\mathbf{Y}_{\mathbf{es}}$ | ı | \mathbf{Yes} |
| Observations | 202474 | 205014 | 202474 | 205014 |
| R-squared | 0.674 | 0.694 | 0.674 | 0.694 |

Acknowledgements

We thank Simone Boldrini, Elena Carletti, Hans Degryse, Mariassunta Giannetti, Luigi Guiso, Vasso Ioannidou, Dmitry Khametshin, Maroje Lang, Simone Manganelli, Jean-Stéphane Mésonnier, Malina Papoutsi, Fabiano Schivardi, Sujiao Zhao and participants to the the 2024 Adam Smith Workshop, the ECB Climate Risk Center Seminar, the 30th Finance Forum, the Banco de Espana–CEMFI–UIMP Conference, the 9th annual meeting of the Task-force on Banking Analysis for Monetary Policy of the ESCB, the Inaugural Conference of the ESCB Research Network on Challenges for Monetary Policy Transmission in a Changing World ("ChaMP"), the ECB–OECD Workshop "Navigating the path to net zero", and the workshop "Climate Finance in a World of Fundamental Uncertanty" at Bocconi University, the 18th Csef-Igier Symposium on Economics and Institutions, the IMF and Bank of Italy Conference "Embedding Sustainability in Credit Risk Assessment" in Venice, the 13th MoFiR Workshop on Banking and seminar participants at EIEF, Essex University, Bank of Italy, Bank of Portugal and Bundesbank for useful discussions and suggestions. We are grateful to Josè Ramos for excellent research assistance. Research by Marco Pagano has been supported by the Wim Duisenberg Fellowship Program sponsored by the ECB.

Any views expressed are only those of the authors and do not necessarily represent the views of the ECB or the Eurosystem.

Carlo Altavilla

European Central Bank, Frankfurt am Main, Germany; Centre for Economic Policy Research, London, United Kingdom; email: carlo.altavilla@ecb.europa.eu

Miguel Boucinha

European Central Bank, Frankfurt am Main, Germany; email: miguel.boucinha@ecb.europa.eu

Marco Pagano

University of Naples Federico II, Naples, Italy; Centre for Studies in Economics and Finance, Naples, Italy; Einaudi Institute for Economics and Finance, Rome, Italy; Centre for Economic Policy Research, London, United Kingdom; European Corporate Governance Institute, Brussels, Belgium; email: pagano56@gmail.com

Andrea Polo

Luiss University, Rome, Italy; Einaudi Institute for Economics and Finance, Rome, Italy; Centre for Economic Policy Research, London, United Kingdom; European Corporate Governance Institute, Brussels, Belgium; email: andreapolo@gmail.com

© European Central Bank, 2024

Postal address 60640 Frankfurt am Main, Germany Telephone +49 69 1344 0 Website www.ecb.europa.eu

All rights reserved. Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the ECB or the authors.

This paper can be downloaded without charge from www.ecb.europa.eu, from the Social Science Research Network electronic library or from RePEc: Research Papers in Economics. Information on all of the papers published in the ECB Working Paper Series can be found on the ECB's website.

| PDF ISBN 978-92-899-6826-3 ISSN 1725 | doi:10.2866/70527 QB-/ |
|--------------------------------------|------------------------|
|--------------------------------------|------------------------|

QB-AR-24-086-EN-N